Author's response to reviews

Title: Successful treatment of paraneoplastic hypercalcaemia in a patient with giant condyloma acuminatum: a case report

Authors:

Thomas Linnemann (Thomas.Linnemann@klinikumbielefeld.de)
Frauke Müller (Frauke.Mueller@klinikumbielefeld.de)
Mathias Löhncrt (Mathias.Loehnert@klinikumbielefeld.de)
Peter Hirnle (Peter.Hirnle@klinikumbielefeld.de)
Martin Görner (Martin.Goerner@klinikumbielefeld.de)

Version: 6 Date: 9 August 2013

Author's response to reviews: see over
Title: Successful treatment of paraneoplastic hypercalcaemia in a patient with giant condyloma acuminatum: a case report

Authors:

Thomas Linnemann (Thomas.Linnemann@gmx.de)
Frauke Müller (Frauke.Mueller@klinikumbielefeld.de)
Mathias Löhner (Mathias.loehner@klinikumbielefeld.de)
Peter Hirnle (Peter.Hirnle@klinikumbielefeld.de)
Martin Görner (Martin.Goerner@klinikumbielefeld.de)

Version: 6 Date: 9 august .2013

Author’s response to reviews: see over
Dear editors,

thank you for your last email. We have spent two days in revising our text with a native speaker from Scotland. He’s a teacher, who is working for the british army in Germany. Did you get our last three-pages cover letter with our revisions? With his help we have changed most of the passages to ameliorate this text. Finally, he was able to retrace our passages. Could you please mention the mistakes in our text we have skipped unfortunately?

We removed the yellow headlined mistakes from our former version.

With kind regards

Thomas Linnemann